Author's response to reviews

Title: STAT1-dependent expression of energy metabolic pathways links tumor growth and radioresistance to the Warburg effect

Authors:

Sean P Pitroda (spitroda@uchicago.edu)
Bassam T Wakim (bwakim@mcw.edu)
Ravi F Sood (rfsood@uchicago.edu)
Mara G Beveridge (beveridge@uchicago.edu)
Michael A Beckett (mbeckett@uchicago.edu)
Dhara M MacDermed (dmacdermed@radonc.bsd.uchicago.edu)
Ralph R Weichselbaum (rrw@radonc.bsd.uchicago.edu)
Nikolai N Khodarev (n-khodarev@uchicago.edu)

Version: 2 Date: 2 October 2009

Author's response to reviews:

Dear Dr. Marshall and Ms. Qazi:

The authors are grateful for the acceptance of our article and the suggestion to publish it in BMC Medicine. It is an honor for us to accept this offer.
Please, let us know how we should proceed.

As requested, we added the GEO accession number for our microarray data in the Results section (page 10, last line). Please, see attached text.

Thank you again.

Sincerely,

Nikolai N. Khodarev
Sean P. Pitroda